Food Policy xxx (xxxx) xxxx



Contents lists available at ScienceDirect

Food Policy



journal homepage: www.elsevier.com/locate/foodpol

Local food prices and the purchasing power of SNAP benefits

Garret Christensen^{a,*}, Erin Todd Bronchetti^b

^a Economist, US Census Bureau, United States

^b Swarthmore College, Swarthmore, PA, United States

ARTICLE INFO

Keywords: Supplemental Nutrition Assistance Program SNAP generosity Food stamps Thrifty Food Plan

ABSTRACT

While the nominal value of Supplemental Nutrition Assistance Program (SNAP) benefits is fixed across states (except for Hawaii and Alaska), variation in food prices across the U.S. is dramatic. We provide new evidence describing geographic variation in the purchasing power of SNAP benefits, measured by the extent to which SNAP-recipient households are able to afford the Thrifty Food Plan (TFP), the U.S. Department of Agriculture (USDA) food plan on which legislated SNAP benefit levels are based. For more than one-quarter of SNAP households, SNAP benefits are too low to cover the cost of the TFP at the primary stores where they report shopping. SNAP purchasing power increases somewhat as we assume households can travel farther to shop and increases much more with the assumed ability to identify and travel to the *lowest-cost* store in a given area. It is unlikely, however, that SNAP households are sufficiently informed and mobile to shop at the lowest-cost store in a large (e.g., 10 to 20-mile) geographic area. We demonstrate that aggregate dollar shortfalls for SNAP households in low-cost areas to those in high-cost areas, e.g., by indexing SNAP benefits to local food prices.

1. Introduction

The Supplemental Nutritional Assistance Program (SNAP, formerly known as Food Stamps), is one of the largest government assistance programs for the poor in the United States, with nearly 1 in every 8 Americans participating in the program and benefit payments exceeding 65 billion dollars in 2018. A substantial body of literature has demonstrated that SNAP leads to short- and long-run improvements in outcomes like health, education, and economic self-sufficiency, particularly for those who receive benefits as children, and significantly reduces food insecurity in recipient households.¹ Despite the program's successes, rates of food insecurity among SNAP recipients remain high, at over 50 percent (Coleman-Jensen et al., 2012), indicating that many SNAP households may be unable to afford a nutritious diet.

Because SNAP benefit levels are determined nationally and fixed across states (except for Alaska and Hawaii), differences in local food prices across the country can generate wide variation in the *real* value –

or purchasing power – of SNAP benefits. Using data on food prices across 35 market groups in the U.S., Todd et al. (2010) and Todd et al. (2011) demonstrate dramatic variation in regional food prices, with prices ranging from 70 to 90 percent of the national average at the low end to 120–140 percent at the high end.² Gregory and Coleman-Jensen (2013) confirm that households in market areas with higher food prices are more likely to be food insecure.

This paper provides new evidence describing geographic variation in the purchasing power of SNAP benefits using a unique, nationally representative data set that allows us to match detailed information on SNAP households to the *local* food prices these households face, including at the stores where they actually shop. We measure SNAP purchasing power by calculating the fraction of SNAP-recipient households that are able to afford the local cost of the Thrifty Food Plan (TFP), a food plan constructed by the U.S. Department of Agriculture (USDA) to represent a nutritious diet at a minimal cost. Weighing SNAP benefits against the local cost of the TFP is sensible because the TFP

https://doi.org/10.1016/j.foodpol.2020.101937

^{*} Corresponding author.

E-mail addresses: garret.s.christensen@census.gov (G. Christensen), ebronch1@swarthmore.edu (E.T. Bronchetti).

¹ For a recent, comprehensive review of the SNAP program and its impacts, see Hoynes and Schanzenbach (2016). Recent evidence of the positive effect of SNAP on health outcomes can be found in East (2016); Gregory and Deb (2015); Hoynes, Schanzenbach, and Almond (2016); Schmeiser (2012); and Almond, Hoynes, and Schanzenbach (2011), among others. SNAP's beneficial impact on food insecurity is documented in recent work by Schmidt, Shore-Sheppard, and Watson (2016); Shaefer and Gutierrez (2013); Mykerezi and Mills (2010); Nord and Golla (2009); Yen, Andrews, Chen, and Eastwood (2008), and reviewed in Gregory, Rabbitt, and Ribar (2015).

 $^{^{2}}$ Both studies use data from the Quarterly Food at Home Price Database (QFAHPD).

Received 30 May 2019; Received in revised form 6 June 2020; Accepted 15 June 2020 0306-9192/ Published by Elsevier Ltd.

serves as the basis for legislated SNAP benefits,³ and because the TFP is a standardized index that is not influenced by any (endogenous) changes in food choices that recipients make in response to higher prices.

Using data from the National Household Food Acquisition and Purchase Survey (FoodsAPS) and FoodAPS-Geography Component (FoodAPS-GC) data sets, we are able to account for variation in local food prices at a much tighter geographic level than has been possible in prior research. Rather than rely on regional food price indices, we link households to multiple local measures of the cost of the TFP they face, using prices from the stores where they are likely able to shop (e.g., stores within given distances) and from the stores at which they report shopping. We then compare several measures of the local cost of the TFP to the financial resources available to the household to spend on food (either SNAP benefits plus 30 percent of net income, or the maximum SNAP benefit for the household's family size).

One key finding is that many SNAP households are unable to afford the TFP at their local stores. For more than one-quarter of SNAP households, SNAP benefits are too low to cover the cost of the TFP at the primary stores where they report shopping. We examine the extent to which SNAP purchasing power increases as households are assumed to be able to travel longer distances to shop. The fraction of recipients who can afford the TFP is fairly stable across different geographic proximity measures but increases slightly as we allow for households traveling farther. For instance, 74 percent of SNAP recipient shoppers can afford the TFP at the median-cost store within 2.5 miles, 75 percent can afford the TFP at the median-cost store in a 20-mile radius, and 77 percent can afford the TFP at the median-cost store in their county.

On the other hand, if one assumes SNAP-recipient households can identify and shop at the store with the lowest TFP cost in their area, the fraction that can afford the TFP is much higher. Of course, the assumption that households can identify and travel to the area store with the lowest TFP cost ignores the potentially high cost of such travel for low-income SNAP shoppers. Even if shoppers were perfectly informed about area stores' prices, traveling to the lowest-cost store may involve significant costs (both financial and time costs) that could outweigh their savings on food. These costs are likely to be higher for the 33 percent of FoodAPS SNAP recipients who do not have a car, or for the 86 percent living in metropolitan areas that often have higher prices.

We also consider the average dollar shortfalls for SNAP households who cannot afford the TFP (at mean or median area prices). For the 20–25 percent of SNAP households for whom benefits are found to be insufficient, we compute the average difference between the local cost of the TFP and the resources available to the household to spend on food. These households face sizeable average shortfalls of approximately \$160 per month, compared to approximately \$230 in monthly benefits received and approximately \$560 in average monthly income.

An important takeaway from these results is that in the aggregate, the dollar shortfalls for SNAP households who cannot afford the TFP could be completely eliminated by redistributing some benefits from households whose SNAP benefits are *more than* sufficient to afford the TFP. That is, policy makers could make the TFP affordable for 100 percent of SNAP households without any additional benefit expenditures by adjusting SNAP benefits for geographic variation in food prices. In the discussion that follows, we explore how the government might go about such an adjustment using existing/available data sets on area food prices, and estimate the aggregate costs of increasing SNAP benefits for those in high-cost counties based on county-level TFP prices.

Our study contributes to the literature on food assistance and food security in a few key ways. We provide some of the first evidence on the purchasing power of SNAP benefits relative to local food prices and show the extent to which SNAP benefits go further when households are able to travel longer distances to shop, or to identify the lowest-cost stores in their areas. This evidence adds to our growing understanding of how and where SNAP households shop. Ver Ploeg et al. (2015) show that SNAP shoppers travel farther (an average of 3.4 miles) to shop at their primary store than to the closest area store that accepts SNAP (on average, 2 miles away from home). While there are a number of possible reasons for traveling beyond the closest store, one possibility is that SNAP households are trying to make their benefits go as far as possible by choosing stores with low prices. However, our evidence suggests that many SNAP households' primary stores are not the local stores at which the TFP costs the least. This may be because SNAP households are unable to identify the minimum-TFP-cost store, are buying different bundles than the TFP (and perhaps choosing their primary stores based on the prices of the items they are buying), or are choosing stores based on other factors altogether (e.g., selection or proximity to work or public transportation).

More broadly, our analysis relates to other work on how high or low local prices can affect the real value of nominally equitable policies. Çakır et al. (2018); and Leibtag & Kumcu (2011) demonstrate that substantial regional variation in produce prices is likely to affect the buying power and nutritional benefits of the Special Supplemental Nutrition Program for Women, Infants, and Children (WIC) fixed-value voucher for purchasing fruit and vegetables. Beyond the literature on food assistance and nutrition programs, Albouy (2009) shows that the nominally equal rates of federal taxation across the country result in substantial penalties in high-wage urban areas and subsidies for lowwage rural areas.

Finally, our work highlights that geographic variation in SNAP purchasing power can provide a source of plausibly exogenous variation in the (real) generosity of benefits. Whereas prior quasi-experimental evaluations of SNAP have focused on the rollout of the program or on changes in eligibility, in related work we use variation in SNAP purchasing power to estimate the impacts of changes in legislated SNAP benefit generosity on child health (Bronchetti, Christensen, and Hoynes 2019), and on the nutritional content of SNAP recipients' diets (Bronchetti, Christensen, and Hansen, 2017). Further research using this approach could add valuable, policy-relevant evidence on the program's short- and long-run impacts.

The paper proceeds as follows. Section 2 provides background on the Thrifty Food Plan (TFP) and describes the FoodAPS data and our methods, including how we construct measures of the local cost of the TFP. Section 3 presents our main results and discusses some of the limitations of our analysis. Section 4 discusses the policy implications of our findings, and Section 5 concludes.

2. Methodology

This paper provides new evidence on the purchasing power of SNAP benefits by assessing SNAP households' ability to buy a specific bundle of food – the USDA's Thrifty Food Plan (TFP) – given local food prices. To do so, we match information on SNAP households in the National Household Food Acquisition and Purchase Survey (FoodAPS) to information on the food retail environment in each household's surrounding area from the FoodAPS Geography Component (FoodAPS-GC).

2.1. The Thrifty food Plan (TFP) and local food prices

We use the TFP to measure the local cost of food for low-income households because the TFP serves as the basis for legislated SNAP benefit allotments and because it provides a standardized bundle of

³ The national average price of the TFP is used as the basis for legislated maximum SNAP benefit levels. Household benefit levels are then set such that households should be able to purchase the TFP with benefits plus 30 percent of their net income (i.e., gross income minus allowed deductions). Said differently, a household's SNAP entitlement is the maximum benefit for its size minus 30 percent of its net income.

foods, unaffected by demand responses to local prices. The TFP was constructed by the USDA to specify a food plan that would be as close as possible to the desired consumption bundle of low-income households, subject to the constraints that the bundle was affordable, provided sufficient food energy, and reflected a nutritious diet (Wilde and Llobrera 2009).⁴ The end result is a set of market baskets – one for each of 15 age-gender groups – that specify the quantities of 59 food categories that could be consumed to obtain a nutritious diet at minimal cost. SNAP maximum benefits are based on the cost of the TFP for a family of four, with two adults and two children (ages 6–8 and 9–11), and then adjusted for family size. In 2018 the national average cost of the TFP for such a family was \$148.70 per week.

The first step in our research is to link each SNAP household to measures of what it would cost to purchase the TFP from local stores. The FoodAPS-GC contains retail food price data compiled by researchers at the University of Illinois and the University of Florida (see Gundersen et al., 2016). The researchers used Information Resources, Inc. (IRI) scanner data on UPC-level sales to construct the price-perpound for each of the TFP food categories, and then computed weekly store-level basket prices as the sum of these prices times the quantities specified in the TFP. We summarize this price data in Table 1. Overall, stores have an average of 6900 UPC items, with 46 percent of stores selling all of whole grains, dark green vegetables, and whole fruit. Restricting to stores with more variety, 29 percent of included stores have items in at least 28 of the 29 TFP food categories (hereafter "full TFP stores"), with an average of 16,800 UPC items.

A concern here is that not all stores are in the IRI data, and not all IRI data was made available to researchers.⁵ Comparing the IRI in FoodAPS to TDLinx stores (the largest national database of stores, compiled by the Nielsen Corporation) to assess coverage of stores by IRI, Fan et al. (2018) show that IRI data covers 90 percent of club stores, mass merchandisers, dollar stores, and drug stores; 74 percent of grocery stores; and 53 percent of convenience stores. While these coverage rates are reasonably high, some large chains that participate in the IRI (including, e.g., CVS, Kroger, Safeway, Publix, and Walmart) only provide aggregate price data for regional marketing areas (RMA), not store-level prices (see Muth et al., 2016).⁶ DellaVigna and Gentzkow (2019) show that chains set uniform or nearly uniform prices across stores, so we are not especially concerned about this as a potential bias to our estimates. Nevertheless, it is important to remember that even if pricing is uniform across stores in these chains, the geographic variation in food prices captured by our measures of the local TFP cost will reflect both the set of stores that is available to a household within a certain distance and the prices at the non-RMA stores in that set. We note, also, that our measures of the local cost of the TFP are based on stores that are represented in the IRI data shared with

FoodAPS and not *all* stores at which respondents might shop, and that some households may not have an IRI store in their immediate area. We return to this issue below.

The data set contains two cost variables, basket_price and low_basket price, which can be used to reflect the cost of the TFP at a store. The first takes the median price-per-pound for each TFP category, multiplies that price by the quantity (in pounds) prescribed for the TFP, and sums across TFP categories. The latter makes the same calculation, but computes the median price-per-pound only among items in the lowest decile of prices for that TFP category. We employ the latter measure throughout our analysis for three reasons. First, Gundersen et al. (2016) note that their basket price calculations may overestimate the cost of the TFP because they are based on a store's sales of all food items in each TFP category, including goods that low-income households may be less likely to purchase. Using the low_basket_price index helps to mitigate this because it focuses on low-priced items within each category. Second, the assumption that SNAP households purchase lower priced items within food categories seems reasonable. Finally, using the lower of the two estimates of TFP cost will tend to bias us toward more conservative (i.e., higher) estimates of SNAP purchasing power.

Fig. 1 demonstrates substantial geographic variation in this measure of TFP cost, both across counties and across smaller areas (defined by a 3.4-mile radius around the census block group centroid).⁷ Reassuringly, our estimates of median TFP costs center around \$140, which is similar to the published national average weekly TFP cost estimate for this time period of \$144.⁸

We match households to local basket prices by both location and survey week. For subjects surveyed in January 2013, after store price data became unavailable, we assign the basket price from the final week of store price data (from December 2012).⁹ We then multiply the weekly cost by 4.3 to obtain a monthly figure for comparison with monthly SNAP benefits.

The estimated cost of the TFP also varies according to a household's size and age-gender composition. The TFP cost used for determining SNAP benefit levels, as well as the estimates in the FoodAPS-GC, are for a family of four, with two adults (male and female, both ages 19–50) and two children (age 6–8 and age 9–11). We adjust for family size using the standard adjustment suggested by the USDA Center for Nutrition Policy and Promotion (CNPP)¹⁰ but we do not fully disaggregate the TFP cost estimates in the FoodAPS-GC to account for the

⁴ For each of 15 age-gender groups and 59 food categories, the USDA used food intake data from the National Health and Nutrition Examination Survey (NHANES) to compute average consumption among low-income households. The USDA then used data from the ACNielsen Homescan Panel to create a database of food prices and assign prices to each of the 59 food groups. Nutrient constraints were taken from the *Dietary Guidelines for Americans* and the Institute of Medicine, and food category quantity constraints were based on the USDA's *MyPyramid* recommendations. See Carlson et al. (2007) for more details.

⁵ IRI data generally includes a "census" component of companies that provide data for *all* locations of their stores, and a "sample" component that is a representative sample. IRI does not share its proprietary sample component, so the FoodAPS data contains only the census component. For more information, see Muth et al. (2016, p. 20).

⁶ Chains providing this data are referred to as Regional Market Area (RMA) stores. Overall, 32 percent of stores are RMA stores, while 56 percent of stores with 28 or more TFP categories are RMA stores. For further information on the coverage of stores in the FoodAPS-GC data, see <u>https://www.ers.usda.gov/media/8582/priceindexdata.pdf</u>, p. 10.

⁷ To reduce the risk of disclosing confidential FoodAPS information, the USDA Economic Research Service (ERS) prevents us from showing a map or geographic distribution of these prices, as the FoodAPS-GC data only contain prices from the primary sampling units and neighboring counties in which FoodAPS households reside. FoodAPS prevents disclosure of data at the county level or a finer geographic level.

⁸ The basket price measure might underestimate the true cost of the TFP at a store because IRI stores that do not sell particular items prescribed by the TFP do not include a price estimate for that item or food category. This would tend to bias our estimates of SNAP purchasing power upward, since basket prices will be the sum of fewer positive values. Column 3 of Table 4 shows that results are similar when we use only stores with near-complete TFP baskets. See <u>https://www.cnpp.usda.gov/USDAFoodPlansCostofFood/reports</u> for TFP measures over time.

⁹ Ignoring the week of basket price data collection completely, and instead assigning respondents to the average TFP price over the entire survey period yields nearly identical estimates.

¹⁰ This adjustment is described in the monthly USDA Cost of Food report, as follows: "The costs given are for individuals in 4-person families. For individuals in other size families, the following adjustments are suggested: 1-person—add 20 percent; 2-person—add 10 percent; 3-person—add 5 percent; 4-person—no adjustment; 5- or 6-person—subtract 5 percent; 7- (or more) person—subtract 10 percent. To calculate overall household food costs, (1) adjust food costs for each person in household and then (2) sum these adjusted food costs." See https://www.cnpp.usda.gov/sites/default/files/CostofFoodSep2016.pdf, for more information.

Table 1

Summary of IRI basket price data.

	All	Stores with $> = 28$ TFP Categories	Stores with 29 TFP Categories
Total observations	1,186,954	381,590	231,246
Average weeks observed per store	51	57	64
Stores (N)	23,147	6,659	3,623
Stores with a price every week (N)	21,756	6,474	3,614
RMA stores (N)	7,409	3,744	3,497
Stores with a price every week (%)	94%	97%	100%
RMA stores (%)	32%	56%	97%
TFP categories per store	23.92	28.66	29.00
UPC count	6,891.5	16,831.6	23,768.2
UPC count: whole grain	13.4	45.2	65.3
UPC count: dark green veg	14.3	47.3	64.1
UPC count: whole fruit	106.2	304.6	427.7
Stores with any whole grain (%)	58%	100%	100%
Stores with any dark green veg (%)	54%	100%	100%
Stores with any whole fruit (%)	96%	100%	100%
Stores with whole grain, dark green veg, and whole fruit (%)	46%	100%	100%

Notes: Table summarizes the food price data available in the FoodAPS-GC, containing repeated weekly observations of UPC counts, TFP categories, and prices at stores in IRI data. The first column summarizes all stores while the second and third are restricted to stores with a weekly average of at least 28, or all 29, TFP categories available for purchase.



Fig. 1. Geographic variation in FoodAPS-GC low-cost basket prices (Distribution of median low-cost basket prices within county and 3.4-mile radius, for all FoodAPS households and SNAP-recipient households).

age and gender composition of sample households. For example, our TFP cost estimates will be the same for a family with 2 adults and one teenager as for a family with a single mother and two young children, even though these households have different nutritional needs.¹¹

We analyze the purchasing power of SNAP to purchase the TFP using multiple measures of the local TFP cost faced by respondent households, which involve different assumptions about how and where respondents shop:

- basket cost at the primary and alternate stores at which the respondent reports shopping, as well as the average of these two basket costs
- the mean, median, and minimum basket cost in the respondent's county
- the mean, median, and minimum basket cost at stores within an X-mile radius of the respondent's census block group centroid (where X = 20, 10, 5, 3.4, 2.5)¹²
- the mean, median, and minimum basket cost at the X stores nearest to the respondent's census block group centroid (where X = 10, 5, 2, 1).

¹¹ This simplification is unlikely to make a significant difference—among FoodAPS SNAP families with four people, the average number of children is 1.85, close to the 2 assumed by the formula. The cost of the TFP for a child is approximately 90% of the cost for an adult woman and 80% of the cost for an adult man.

¹² We choose 3.4 miles here because that is the population weighted average of the straight-line distance to shoppers' primary store.

Given that some local stores may not be in the FoodAPS-GC data, either because they do not participate in IRI or are not in the subset of IRI stores shared with end users, it is encouraging that even at the smallest geographic level (a radius of 2.5 miles), we are able to link over 80 percent of all SNAP households to a local TFP cost estimate. Of course, that percentage rises as we use larger areas to estimate the local TFP cost faced by the household. We also investigate the characteristics of respondents whom we can link to a store and compare them to characteristics of those without IRI-covered stores near them. SNAP households for whom we cannot observe a local TFP cost estimate tend to be older and less likely to live in a metro area, but are otherwise similar to those whom we can match to a local store.¹³ Finally, in the Appendix we demonstrate that our results are qualitatively very similar if we use a consistent sample that only includes SNAP households for whom we can observe a TFP cost estimate in the smallest radius we describe above (2.5 miles).14

2.2. Data on SNAP households and food shopping

The FoodAPS is a nationally representative survey of nearly 5000 households, conducted by the USDA's Economic Research Service (ERS) between April 2012 and mid-January 2013. FoodAPS data include detailed information on the food purchases and acquisitions of nearly 5000 households, as well as information on their demographic characteristics, income and employment, and SNAP participation. An advantage of the FoodAPS data is that survey responses are matched to SNAP administrative records so that self-reports of SNAP participation and benefit receipt can be confirmed (see Clay et al., 2016). We focus on the sample of FoodAPS respondent households who received SNAP benefits in the past month.¹⁵ These SNAP participant households are oversampled by the survey: Of the 4826 households in the dataset, 1581 (33 percent) were receiving SNAP benefits at the time of interview.

While the primary focus of the survey was a detailed tracking of all food acquired by the household (both quantities and expenditures) from all sources over a one-week period, the data set's Geography Component (FoodAPS-GC) contains detailed information on the food retail environment in each household's surrounding area. Using these data, we are able to match households to stores (and prices) at the level of the census block group, rather than to stores within a wider geographic area, as in prior research. Geographic identifiers are masked in the public data, but they are made available to researchers on a restricted-use basis.¹⁶

It is worth considering how reliably the FoodAPS measures outcomes related to food spending and shopping, SNAP participation, and income. Clay et al. (2016) compare the FoodAPS to data from other national surveys that gather information on these topics. They document that FoodAPS finds a five percent greater amount of spending on food than the Bureau of Labor Statistics' (BLS) Consumer Expenditure Survey (although there is no difference for households with children), and significantly more food insecurity than in either the National Health Interview Survey (NHIS) or the Current Population Survey-Food Security Supplement. However, the main variables of interest to our analysis are SNAP participation and income. Compared to data from the Survey of Income and Program Participation (SIPP), FoodAPS estimates a nearly identical rate of SNAP participation (13.6 percent). For SNAP participating households, FoodAPS estimates somewhat higher average incomes than does the SIPP. Because FoodAPS cannot precisely measure the SNAP unit(s) within the household, it may overestimate income for each SNAP household (e.g., a household containing two SNAP units would be treated as a single SNAP unit, with all household income attributed to it). In Section 2.3, we explain that to the extent that FoodAPS overestimates income for SNAP households, this is likely to bias our estimates of SNAP purchasing power upward (i.e., toward 100 percent).

2.3. Measuring resources available to the household to purchase food

To analyze the purchasing power of SNAP benefits, we compare each household's resources for purchasing food to the local TFP cost estimates described above. We describe SNAP purchasing power as the fraction of households who can afford the TFP based on two measures of the resources available for purchasing food: (1) SNAP benefits received plus 30 percent of net income and (2) the maximum legislated SNAP benefit for the household's size.

We estimate a household's net income by subtracting from its reported gross income all the SNAP-allowed deductions for costs associated with housing, earnings, dependent care, medical expenses and child support payments. We use household-level and person-level data to estimate the amount of these deductions.¹⁷ Of particular importance is the excess shelter deduction, not only because of its size, but because it is the primary deduction that would help to offset geographic variation in other prices (i.e., of non-food items). The excess shelter deduction allows households to deduct any shelter expenses (e.g., mortgage or rent, plus utilities) that exceed 50 percent of their income after all other deductions have been made, up to a cap (\$459 in 2012).

We use 30 percent of income because SNAP benefit amounts are designed with the assumption that recipient households spend approximately 30 percent of their cash resources on food. Accordingly, a family's SNAP benefit is determined by subtracting 30 percent of the family's net income from the maximum legislated benefit, which is set equal to the national average cost of the TFP. When the deductions described above reduce a household's net income to zero, the household receives the maximum benefit. Approximately 10 percent of SNAP recipient households in our sample are determined to have no net income and thus receive the maximum benefit.

Given how SNAP benefit levels are calculated, our two measures of household resources for food spending (benefits plus 30 percent of net income and the SNAP maximum benefit level) would be identical with perfect reporting and program administration, and if we correctly calculate the deductions from gross income.¹⁸ In practice, however, there

¹³ See Appendix Table 1.

¹⁴ Please see Appendix Tables 2, 3, 4, and 5 for tables constructed analogously to Tables 2 through 5 in the main paper.

¹⁵ The household interview file contains variables indicating SNAP participation as reported in the initial interview (SNAPNOWREPORT) and reported SNAP participation status that is revised per the match to administrative data (SNAPNOWHH). We use SNAPNOWHH to define the sample of SNAP recipient households. See page 20 of the FoodAPS documentation at https://www.ers.usda.gov/media/8804/0_foodaps-user-guide-puf.pdf.

¹⁶ See <u>http://www.ers.usda.gov/data-products/foodaps-national-household-food-acquisition-and-purchase-survey.aspx</u>. Due to data access restrictions, we are unable to share these data; however, the USDA has made available a public use data set without geographic identifiers.

¹⁷ The full list of deductions can be found at <u>https://www.fns.usda.gov/snap/</u> <u>eligibility#What deductions are allowed in SNAP?</u> deductions are allowed in <u>SNAP?</u> FoodAPS asks respondents about many of these, which we use to calculate net income: medical expenses for the elderly (EXPPOPMEDICAL60), rent/mortgage (EXPRENTMRTG), heating and cooking fuel (EXPHEATFUEL), electricity (EXPELECTRIC), property taxes (EXPPROPTAX), home insurance (EXPHOMEINS), child care (EXPCHILDCARE), and child support owed (EXP-CHILDSUPP).

¹⁸ Another option in lieu of calculating net income from reported gross income and deductions would be to use the data on benefits and family size to back out net income. In theory this should yield results that are identical to those we find using maximum benefits for a given family size, and the two measures are likely to be very similar in practice. Differences between these three measures arise exclusively from misreporting. We prefer to use the measure of net income calculated from gross income and deductions (which requires the most self-reported data) and the SNAP maximum benefit entitlement (requiring the least self-reported data) to cover the full range of possibilities.

are small but meaningful differences in the results for these two measures. We describe these further below.

In addition to calculating the fraction of households that can afford the local cost of the TFP, we also compute average dollar shortfalls for households for whom SNAP benefits are found to be insufficient to purchase the TFP. Specifically, we calculate the average difference between the cost of the TFP and the household's SNAP benefits plus 30 percent of net income (or the maximum SNAP benefit). We are particularly interested in the distribution of these differences, which sheds light on the feasibility of some policy options, like the adjusting SNAP benefits for local food prices (i.e., redistributing from low food price areas to higher food price areas).

3. Results

3.1. SNAP purchasing power and distance to shop

Table 2 displays evidence on the purchasing power of SNAP benefits, measured as the fraction of SNAP households who can afford the local cost of the TFP. In the top panel, we show that only 77 percent of SNAP households in the FoodAPS data find their SNAP benefits plus 30 percent of net income to be sufficient to afford the national average cost of the TFP (\$145). Because maximum SNAP allotments are set according to the national average TFP cost, all households can afford the statutory TFP cost under the maximum benefit.

More interesting is the share of SNAP households who can afford the TFP at their *local* stores, also displayed in the top panel. Measuring household resources for food as SNAP benefits plus 30 percent of net income, we find that 76 percent of households could afford the TFP at the store located nearest to them, and only 73 percent could afford the TFP at the primary store at which they shop. If SNAP households are assumed to receive the maximum SNAP benefit for their size, these fractions are notably lower, at 68 percent and 63 percent, respectively. These differences may be a result of our overestimating net income (perhaps by underestimating the value of allowed deductions from gross income). If this is the case, our estimates in the first column paint too optimistic a picture of the ability of SNAP households to afford the TFP.¹⁹

Next, we use TFP cost estimates from increasingly local geographic regions-from the national average and Census region-level average TFP costs down to the TFP cost at stores within a 2.5-mile radius. The table also shows results for the 10, 5, 2, and 1-nearest stores. For each set of stores, we compute the fraction of households who can afford the mean, median, and minimum of TFP prices (panels B, C, and D, respectively). We note that the sample size decreases as the TFP cost measure becomes increasingly local because, e.g., not all households can be linked with a TFP cost within 2.5 miles. This could be because there is no store within 2.5 miles, or it could be because stores within that radius are not IRI stores (and therefore, are not observed in the FoodAPS-GC TFP price data). Appendix Table 2 demonstrates that the results are very similar when we limit the analysis to a consistent sample by including only households for whom we have a local TFP cost measure for all of the proximity measures we consider (i.e. households who have an IRI store within 2.5 miles).

Irrespective of how tightly we define the geographic area in which

Food Policy xxx (xxxx) xxxx

Table 2

Fraction of SNAP recipient households that can afford the local cost of the TFP.

	(1) SNAP Benefits + 30% Net Income Exceeds Local TFP Cost?		(2) Maximum SNAP Benefit Exceeds Local TFP Cost?	
Measure of TFP Cost	% of Households	Sample Size	% of Households	Sample Size
Panel A.				
Statutory TFP cost (\$145)	77%	1444	100%	1581
Mean TFP cost of all	68%	1444	0%	1581
stores in IRI data (\$156)				
TFP cost, nearest store to	76%	1290	68%	1414
respondent				
TFP cost, primary store where respondent reports shopping	73%	723	63%	798
TFP Cost, alternate store	77%	504	70%	552
where respondent reports				
shopping				
Mean TFP cost, primary and alternate stores	74%	985	69%	1082
TFP cost, store where	73%	629	63%	689
respondent shopped most				
Panel B. Mean TFP cost at sto	res within respon	ident's:		
Census region	72%	1444	74%	1581
State	75%	1444	76%	1581
County	74%	1436	76%	1572
20-mile radius	74%	1338	73%	1464
10-mile radius	73%	1311	74%	1433
5-mile radius	71%	1224	74%	1338
3.4-mile radius	72%	1174	74%	1281
2.5-mile radius	71%	1123	72%	1225
10 nearest stores	73%	1338	79%	1464
5 nearest stores	71%	1332	72%	1458
2 nearest stores	73%	1332	72%	1458
Panel C. Median TFP cost at s	tores within resp	ondent's:		
Census region	75%	1444	83%	1581
State	75%	1444	76%	1581
County	77%	1436	75%	1572
20-mile radius	75%	1338	73%	1464
10-mile radius	75%	1311	73%	1433
5-mile radius	74%	1224	72%	1338
3.4-mile radius	74%	1174	74%	1281
2.5-mile radius	74%	1123	72%	1225
10 nearest stores	75%	1338	77%	1464
5 nearest stores	74%	1332	71%	1458
Panel D Minimum TFP cost a	stores within re	spondent's:		
Census region	100%	1444	100%	1581
State	99%	1444	100%	1581
County	94%	1436	100%	1572
20-mile radius	94%	1338	100%	1464
10-mile radius	92%	1311	100%	1433
5-mile radius	90%	1224	99%	1338
3.4-mile radius	89%	1174	100%	1281
2.5-mile radius	89%	1123	99%	1225
10 nearest stores	89%	1338	100%	1464
5 nearest stores	87%	1332	98%	1458
2 nearest stores	81%	1332	85%	1458

Notes: The first and third columns display the survey-weighted fraction of households who can afford the local cost of the TFP. The local cost of the TFP is based on weekly store-level "basket prices" computed from IRI scanner data on UPC-level sales (see Gundersen et al., (2016)). Respondent households are matched to weekly basket prices at IRI stores according to household's census block group and week of interview. Some households do not have an IRI store within the given radius of their home, causing the sample size to fall as we decrease the distance to the respondent's home when defining the "local" cost of the TFP. See Appendix Table 2 for results using consistent samples of households who have non-missing TFP price estimates within the smallest radius (2.5 miles).

¹⁹ Note that each fraction is calculated for the sample of households for which we can calculate the given measure of local TFP cost using FoodAPS-GC data. For example, if the store located nearest to the household store is not an IRIparticipating store, the FoodAPS-GC will not contain price data for that household-store combination, and the household will not be included our sample for that calculation. Sample sizes differ between columns 1 and 2 because income data are missing for some households. Estimates are nearly identical when we restrict the maximum benefits sample to those for whom we can also calculate net income (keeping sample sizes equal across columns 1 and 2). Results available upon request.

households shop, we find that between 71 percent and 79 percent of households can afford to purchase the mean or median TFP cost at *local* stores.²⁰ For example, when the estimated TFP cost is based on prices at stores within 2.5 miles from the block group centroid where a respondent resides, we find that 74 percent of households can afford the median TFP cost with SNAP benefits plus 30 percent of net income. This share only increases by 3 percentage points, to 77 percent, when we compare household resources to the county-level median TFP cost. As above, the fraction that can afford the TFP at median store prices is slightly lower when we compare maximum benefit levels to the estimated TFP cost instead of SNAP benefits plus 30 percent of net income.

In the bottom panel of Table 2, we describe what happens to SNAP purchasing power when one assumes that SNAP recipients can identify and shop at the *lowest* cost store within a given area. Not surprisingly, the fraction of households that can afford the TFP using their SNAP benefits plus 30 percent of net income increases dramatically, ranging from 81 percent when we assume shoppers purchase the TFP at the lowest cost of the two nearest stores, to 94 percent when they purchase the TFP at the minimum-cost store in the entire county. We also note that SNAP households report traveling an average of 3.4 miles to their primary stores, at which 73 percent of them can afford to purchase the TFP. By our calculations, SNAP purchasing power increases meaningfully if these households instead shop at the *lowest-cost* store within that same distance, with 89 percent of them able to afford the TFP.

While more SNAP households can afford the TFP at the minimumcost store in their areas, we caution against too optimistic an interpretation of these results for several reasons. First, recall that we are already imposing the assumption that within any given store, shoppers purchase TFP items with prices in the lowest decile of prices for that TFP category; nonetheless, the fraction of shoppers who can afford this price at the lowest-cost store in their county (using their SNAP benefits plus 30 percent of net income) is still meaningfully lower than 100 percent. Second, given the large size of most counties (the median is over 600 square miles), it seems extremely unlikely that most shoppers are able to identify and travel to such a store. Even if shoppers were to do so (e.g., travel halfway across a median-sized county to shop at the store with lowest TFP cost), they would incur significant travel costs (both financial and time), which may outweigh their savings on food. These costs are likely to be higher for the 33 percent of FoodAPS SNAP recipients who do not have a car, or for SNAP recipient households who live in high-priced, urban areas. While significant savings might be achievable by traveling 10+ miles to the lowest TFP-cost store in the county, the barriers to doing so are likely to be prohibitively high in this population. Finally, our estimates of the minimum TFP cost within a given area may understate the true local cost of the TFP because stores without any foods in a certain TFP category will have an artificially low TFP cost in our data.²¹ We discuss this issue further below and conduct a robustness check using TFP cost measures that come only from stores that have items (and thus, prices) for at least 28 of the 29 TFP categories. Indeed, our estimates of SNAP purchasing power are significantly lower when we do so.

In Table 3 we describe the characteristics of households with high versus low SNAP purchasing power (i.e., for whom the SNAP maximum benefit is sufficient versus insufficient to purchase the TFP at the county median price). Of course, recipient households with low SNAP purchasing power are significantly more likely to live in high food price areas (defined as the 75th percentile of national TFP estimates) and more likely to reside in metropolitan areas. They also have higher average incomes and are more likely to have a college degree,

suggesting that the extra income urban residents tend to earn is not sufficiently large to accommodate the increase in the price of food. Most other measures, including those related to material hardship, do not differ significantly across the two samples, perhaps due to the limited size of the samples. For example, households with low SNAP purchasing power appear somewhat less likely to own a car, suggesting perhaps greater difficulty in traveling to low-cost stores in their areas, but the difference is not statistically different from zero. Similarly, we do not find any significant differences in the unconditional means of variables measuring food security or budgetary strain (i.e., trouble paying bills) for these two groups.

Table 3 also includes information on the type of stores at which household shop. Only two to five percent of SNAP households report that their primary store is neither a supermarket nor a superstore, and differences across purchasing power are not significant. 45–47 percent have a primary store that is a supermarket, while 50–51 percent have a primary store that is a superstore. These stores are likely to have all types of food in the TFP available for purchase, an issue which we return to below.

3.2. Robustness Checks

In Table 4 we present the results of several checks on our main estimates of SNAP purchasing power. First, if take-up of SNAP benefits is endogenous with respect to local prices, and households facing higher local prices are more likely to participate, estimates of SNAP purchasing power among our sample of SNAP-participating households may be too low. To check this, we also conduct the analysis on a sample of FoodAPS households imputed to be eligible for SNAP benefits. FoodAPS uses the Household Survey of Income and Program Participation + (MATH SIPP+) Microsimulation model to impute eligibility for SNAP based on four different simulation models (Leftin, et al. 2014). We use the indicator for eligibility that is generated by their model 4, which differs from the other models in that it allows for multiple SNAP units in a household (whereas models 1 and 2 do not) and adjusts reported net earnings by a factor of 1.4 to approximate gross earnings (while model 3 does not). Model 4 identifies 2405 FoodAPS households as containing an eligible SNAP unit. Eligible households that do not take up SNAP are widely assumed to be better off, on average, than households that do enroll, so readers should interpret these results with caution.

Column 1 of Table 4 shows the fraction of SNAP-eligible households that can afford the local cost of the TFP assuming they receive the maximum SNAP benefit for their size.²² We find that SNAP purchasing power for these households is generally lower than for SNAP recipients (see Table 2), suggesting we are unlikely to be underestimating SNAP purchasing power when we focus on the selected sample of households who have taken up SNAP.

A second concern is that our main results may be based on a sample of disproportionately urban households because many rural areas lack an IRI-participating store and thus, SNAP households in those areas cannot be matched with a local TFP cost. If food prices tend to be higher in urban areas, this may cause us to underestimate SNAP purchasing power in Table 2. In column 2 of Table 4, we explicitly limit the sample to households in that live in metropolitan areas.²³ We note that the sample size decreases only about 10 percent from that in Table 2 (column 2), consistent with our sample of FoodAPS SNAP households

 $^{^{20}}$ When we refer to local stores, we do not include the comparisons to TFP costs within the household's state or Census region.

²¹ Recall that basket price estimates are not scaled or corrected for missing food categories; when a store has no items in a food category the basket price is just a sum of fewer positive terms.

 $^{^{22}}$ We use maximum SNAP benefits as the measure of resources available to the household to avoid confounding any effects of the change in sample with effects of having to use simulated SNAP benefits for households that do not receive them. In results not shown here, we find that the FoodAPS simulated SNAP benefits are much higher for the sample of SNAP-recipient households than the self-reported, administratively confirmed benefit amounts.

²³ Again for the sake of comparison, we use a household's maximum SNAP benefit as the measure of resources available for spending on food.

Table 3

Average characteristics o	f SNAP	households,	by	SNAP	purchasing power.
---------------------------	--------	-------------	----	------	-------------------

	(1) Low SNAP PP (Max SNAP Benefit < Local TFP Cost)	(2) High SNAP PP (Max SNAP Benefit > Local TFP Cost)	(3) p-value
Household size	3.03	2.87	0.39
Household max age	50.88	49.33	0.28
Household min age	27.02	28.13	0.66
Household per capita income	865.38	783.11	0.36
Household income (monthly)	2402.43	1947.86	0.05
Percent of federal poverty line	142.51	124.04	0.11
Household has earned income	0.51	0.53	0.63
Household has car	0.61	0.69	0.24
Travel time to primary store (minutes)	11.21	11.61	0.73
Chooses primary store because of prices	0.64	0.60	0.38
Primary store is small (FoodAPS definition)	0.03	0.05	0.33
Primary store is small (SNAP definition)	0.02	0.05	0.20
Primary store is supermarket	0.47	0.45	0.89
Primary store is superstore	0.51	0.50	0.82
Household member has college degree	0.19	0.09	0.00
Household has elderly member	0.3	0.27	0.36
Number of children in Household	1.09	1.03	0.54
Lives in metro area	0.97	0.83	0.01
High food security	0.34	0.32	0.47
Marginal food security	0.25	0.21	0.22
Low food security	0.24	0.26	0.61
Very low food security	0.17	0.21	0.35
Trouble paying bills	0.29	0.28	0.57
High food price area	0.88	0	0.00
Northeast	0.23	0.09	0.25
Midwest	0.24	0.34	0.34
South	0.33	0.43	0.23
West	0.21	0.14	0.48
Number of observations	382	1190	

Notes: Table shows average characteristics of households for whom resources (defined as maximum benefit for household size) are insufficient/sufficient to purchase the TFP at the county median price. Local cost of the TFP is estimated as described in Table 2. Estimates are survey-weighted; column (3) displays p-value on F-test for equality.

disproportionately living in metropolitan areas. We find that the fraction of households who can afford the local cost of the TFP is, not surprisingly, slightly lower here than in Table 2, but still tends to be around 70–75 percent.

Finally, as we described above, the primary measures of local TFP costs may understate the true cost of the TFP because stores that do not sell *any* items in a particular TFP category will have basket prices that are artificially low. In column 3 we show the results of a robustness check wherein we employ TFP cost measures that come from "full TFP stores"—those that have items (and thus, prices) for at least 28 of the 29 TFP categories.

The results, displayed in column 3 of Table 3, indicate significantly lower SNAP purchasing power for households shopping within local areas. The fraction who can afford the TFP is generally several points lower than indicated by the corresponding estimates in Table 2. For example, now only 54 percent of households can afford the TFP at full TFP stores within a 3.4-mile radius (the average reported distance SNAP households travel to shop), a reduction of 20 percentage points relative to the corresponding estimate in Table 2.

While we attempted in Table 2 to provide conservative estimates of

the fraction of SNAP households who can afford the TFP by keeping the sample of stores at which we estimated TFP prices as large as possible, the more dismal estimates in column 3 of Table 4 may better reflect the true purchasing power of SNAP shoppers at the stores where they actually shop. As demonstrated in Table 3, the vast majority of SNAP households report that their primary store for food purchases is a supermarket or superstore, i.e., a full TFP store.

3.3. Do SNAP households shop at the local store with the lowest Prices?

Our results in Tables 2 and 4 consistently show that the fraction of SNAP households who can afford the TFP is substantially higher if households are assumed to be able to shop at the minimum-TFP cost stores in their local areas. A natural question is to what extent households *do* identify and shop at the area stores with the lowest prices. When asked for their reasons for choosing their primary stores, the reason SNAP households reported most frequently was low prices (59 percent). (The next most frequent reason was proximity.)

We examine directly whether the household's reported primary store is, indeed, the local store at which the TFP costs the least. To do so, we limit the sample of SNAP households to those whose primary store is one of the IRI stores in the FoodAPS-GC, and continue to limit the set of stores to those with items in at least 28 out of 29 TFP categories (so that our TFP price estimates are not artificially low). We find that the fraction whose primary store is also the lowest TFP-cost store in their local (2.5-mile) area is around 47 percent.²⁴ This fraction is somewhat lower, but broadly consistent with the 59 percent whose stated preferences for low prices affected their choice of their primary store.

On the other hand, this also suggests that roughly half of SNAP households are *not* shopping at the lowest TFP-cost stores nearby. This may be because additional factors weigh into households' decisions about where to shop, SNAP households are buying other items than those in the TFP (and choosing stores with low prices for *those* items), and/or these households are choosing primary stores with TFP prices that are also quite low, but not the very lowest in their area. Examining the data directly, we find that households whose primary stores are *not* the lowest TFP-cost stores in their local areas are significantly more likely to list quality of items or a store loyalty program as reasons for their choice of primary store than those who shop at the lowest TFP-cost store, and significantly less likely to report low prices as a reason for their store preference.

3.4. Budget shortfalls

While the result that roughly a quarter of SNAP-recipient households in our sample are unable to purchase the TFP at local prices is striking, the fraction of households who can or cannot afford the TFP is only one measure of SNAP purchasing power. Here, we shed light on the degree to which households with low SNAP purchasing power fall short of being able to afford the TFP.

Table 5 contains estimates of the average dollar shortfall for SNAPrecipient households for whom SNAP purchasing power is too low to afford the local cost of the TFP. This shortfall is calculated as the difference between SNAP benefits plus 30 percent of net income and the local cost of the TFP, or between maximum SNAP benefits and the cost of the TFP. Using benefits plus 30 percent of net income for the measure of resources available to the household, those who are unable to afford the TFP face a shortfall of \$159–174 each month (if facing the mean TFP cost in their area) or \$145–164 each month (if facing the median TFP cost). Measuring with the maximum benefit yields average

²⁴ The analogous fraction of households for whom the store at which they spent the most is also the lowest TFP-cost store within 2.5 miles is very similar, at 48 percent.

G. Christensen and E.T. Bronchetti

Table 4

Fraction of households that can afford the local cost of the TFP: alternative samples and robustness checks.

Measure of TFP Cost% of Household% of Household%%% <th< th=""><th></th><th colspan="2">(1) SNAP Eligible Households Maximum SNAP Benefit Exceeds Local TFP Cost?</th><th colspan="2">(2) Urban SNAP Households Maximum SNAP Benefit Exceeds Local TFP Cost?</th><th colspan="2">(3) Full TFP Stores Maximum SNAP Benefit Exceeds Local TFP Cost?</th></th<>		(1) SNAP Eligible Households Maximum SNAP Benefit Exceeds Local TFP Cost?		(2) Urban SNAP Households Maximum SNAP Benefit Exceeds Local TFP Cost?		(3) Full TFP Stores Maximum SNAP Benefit Exceeds Local TFP Cost?	
Part A Statury TP cots (145) 00% 2405 00% 1284 630 1531 TPP cots, primary store where respondent shops 67% 1222 58% 511 776 395 TPP cots, primary store where respondent shops 67% 1623 66% 971 69% 1059 TPP cots, diremate store where respondent shops 67% 1024 59% 617 69% 1059 Parter cots core where respondent shops 67% 1024 59% 617 69% 151 County 67% 2405 75% 1430 76% 1527 County 69% 2422 71% 1325 66% 157 County 69% 1622 73% 1144 64% 1176 25-mile radius 66% 1692 73% 1306 68% 1271 3-mile radius 66% 1679 71% 1448 57% 1674 10 nearest stores 68% 2227 76% 122	Measure of TFP Cost	% of Households	Sample Size	% of Households	Sample Size	% of Households	Sample Size
Shattory TP cort (\$145)100%205100%1430100%1581TP cort, nearest store to respondent hops67%212858%71063%776TP cort, nearest store to respondent hops67%85266%97169%1059TP cort, nearest store to respondent hops67%16366%97169%1059TP cort, store where respondent hops67%16366%97169%168Panel B. Man TP cort a stores within respondent71%240579%143074%1581County69%234274%142776%157220-mile radius69%242274%142776%157220-mile radius69%242274%142668%12715-mile radius66%189273%119464%1763-mile radius66%128276%132556%13615 narcer stores64%223776%132557%136110 nearest stores78%24278%143076%15615 narcer stores64%223776%132557%15722-maire stores78%24378%143076%157210 nearest stores78%24375%142776%15722-maire stores78%24378%15721572157210 nearest stores78%24375%1428167	Panel A.						
TFP cot, incerest store to respondent6%216866%128463%935TFP cot, infranzy and interast stores6%85268%51171%510Mean TFP cot, alternate store where respondent shope6%85268%97163%689Panel E. Mean TFP cot at stores within respondents107459%61763%689Panel E. Mean TFP cot at stores within respondents71%240575%143074%1581County69%239275%143076%158220 mile radius69%242271%132569%144910 mile radius66%196273%124668%157220 mile radius66%196273%124668%15723 4-mile radius66%196273%124668%15723 4-mile radius66%196273%124668%13725 nearest stores64%223770%132563%167410 nearest stores79%240581%143083%15815 nearest stores64%239572%142664%13722 nearest stores79%240581%143083%15816 mile radius69%240273%142664%13722 nearest stores79%240581%143083%15816 mile radius69%240573%142664	Statutory TFP cost (\$145)	100%	2405	100%	1430	100%	1581
TPF cost, primary store where respondent shop TPF cost, primary and alternate store where respondent shoped man TPF cost, primary and alternate store where respondent shoped man TPF cost, primary and alternate store where respondent shoped man TPF cost, store where respondent shoped man the sponder mater store state164366%97169%150Panel E. Mean TFF cost at stores within responder County71%240570%143074%1581County69%239574%143074%1581County69%239574%142776%157220-nile radius69%204271%130668%13715-nile radius69%204373%114664%12753-4-nile radius66%196273%114857%107410 nearest stores64%227270%132556%13312-anerest stores64%223770%132556%15812-anerest stores64%223770%132556%15812-anerest stores64%223770%132556%15812-anerest stores64%240575%143083%15812-anerest stores74%240575%143083%15812-anerest stores74%240575%143083%15812-anerest stores74%240575%143083%15812-anerest stores74%240575%13	TFP cost, nearest store to respondent	62%	2168	66%	1284	63%	395
TPF Cost, alternate store where respondent shope 65% 852 66% 511 71% 510 Wean TPF cost, primary and alternate stores 63% 104 59% 617 63% 639 Panel E. Mean TPF cost at stores within respondents' 63% 1531 Census region 71% 2405 75% 1430 76% 1581 County 69% 2392 73% 1325 69% 1449 10-mile radius 66% 2189 73% 1246 68% 1371 3-traile radius 66% 1962 70% 1232 56% 1365 3-traile radius 70% 2247 70% 1232 56% 1361 <td>TFP cost, primary store where respondent shops</td> <td>57%</td> <td>1222</td> <td>58%</td> <td>710</td> <td>63%</td> <td>776</td>	TFP cost, primary store where respondent shops	57%	1222	58%	710	63%	776
Mean TPP cost, strong where respondent shopped mos 5% 1643 66% 971 69% 1059 TPT cost, store where respondent shopped mos 5% 1074 59% 617 63% 669 Panel B. Men TPP cost at stores within respondents': 581 Canuty 69% 2395 74% 1430 76% 1531 County 69% 2395 74% 1427 76% 1572 20-mile radius 69% 2189 73% 1306 66% 1371 5-mile radius 67% 2189 73% 1144 64% 1371 2.5-mile radius 63% 1672 73% 1148 57% 1074 2.5-mile radius 63% 2237 70% 1325 56% 1337 2.6 maerst stores 64% 2377 70% 1325 56% 1337 2.6 maerst stores 64% 2377 70% 1325 56% 1351 2.0 mai	TFP Cost, alternate store where respondent shops	65%	852	68%	511	71%	510
TPP cost, store where respondent shopped mot58%107459%61763%69%Panel LP cost at stores within respondent's71%240575%143074%1581County69%240575%143076%157220-mile radius69%242471%132569%147210-mile radius69%242471%132569%15723-mile radius67%204373%124668%12573-mile radius66%166273%119464%11762-mile radius63%123770%132565%133710 marest stores70%24276%132556%13372-mile radius64%23770%132556%15132-mile radius64%23770%132556%15372-mile radius64%23770%132555%15722-mile radius64%24275%143076%15722-mile radius64%24270%132555%15722-mile radius64%166275%143076%15722-mile radius64%162275%143076%15722-mile radius64%162275%143076%15722-mile radius64%162275%132555%13723-mile radius64%16275%1325 <td< td=""><td>Mean TFP cost, primary and alternate stores</td><td>63%</td><td>1643</td><td>66%</td><td>971</td><td>69%</td><td>1059</td></td<>	Mean TFP cost, primary and alternate stores	63%	1643	66%	971	69%	1059
Paral F. Mean TPP cost at stores within respondent: V Cenus region 71% 2405 75% 1430 74% 1581 State 71% 2405 75% 1430 76% 1581 County 69% 2395 74% 1427 76% 1572 20-mile radius 69% 2395 74% 1426 68% 1371 5-mile radius 67% 243 73% 1246 68% 1257 3-s-mile radius 66% 162 75% 1194 64% 1267 3-s-mile radius 66% 162 75% 1252 63% 1371 2-serrest stores 70% 232 70% 1325 56% 1137 2-nearest stores 64% 237 70% 1325 56% 1651 2-nearest stores 79% 2405 81% 1430 76% 1581 5 nearest stores 79% 2426 75% 1430 76%	TFP cost, store where respondent shopped most	58%	1074	59%	617	63%	689
Census region 71% 2405 70% 1430 74% 1581 State 71% 2395 75% 1430 76% 1572 20-mile radius 69% 2242 71% 1325 69% 1449 10-mile radius 66% 2142 71% 1325 69% 1247 5-mile radius 66% 203 73% 1246 68% 1257 3.4-mile radius 66% 162 73% 1194 64% 1267 2.5-mile radius 66% 2237 70% 1325 65% 1137 2.6-meers stores 64% 2237 70% 1325 56% 1137 2.0-meers stores 64% 2237 70% 1325 56% 1137 2.0-meers stores 64% 2237 70% 1325 56% 1581 5.0-meers stores 71% 2405 75% 1430 76% 1581 5.0-meers stores 71%	Panel B. Mean TFP cost at stores within respondent's:						
State 71% 2405 75% 1430 76% 1581 County 69% 2395 74% 1427 76% 1572 20-mile radius 66% 2189 73% 1306 68% 1371 5-mile radius 66% 189 73% 1306 68% 1257 3.4-mile radius 66% 1962 73% 1144 64% 1276 3.4-mile radius 66% 1962 73% 1148 57% 1074 10 nearest stores 64% 2237 70% 1325 63% 1365 2.5-mile radius 64% 2237 70% 1325 56% 1371 2 nearest stores 64% 2237 70% 1325 56% 1361 State 73% 2405 81% 1430 83% 1581 County 71% 2405 75% 1430 76% 1521 20-mile radius 66% 2189	Census region	71%	2405	70%	1430	74%	1581
County 69% 2395 74% 1427 76% 1521 20-mile radius 69% 2242 71% 1325 69% 1449 10-mile radius 66% 1287 73% 1306 69% 1271 5-mile radius 67% 2043 73% 1146 68% 1257 3.4-mile radius 66% 1267 73% 1148 57% 1074 2.5-mile radius 63% 1277 70% 1225 65% 1307 5 nearest stores 64% 2237 70% 1325 57% 1307 2 nearest stores 64% 2242 70% 1325 57% 1581 5 nacrest stores 64% 2242 70% 1430 76% 1581 6 county 73% 2405 72% 1427 75% 1572 20-mile radius 69% 2189 71% 1306 49% 1571 20-mile radius 69% <	State	71%	2405	75%	1430	76%	1581
20-mic radius69%24271%132569%144910-mic radius6%218973%130668%13715-mic radius6%196273%114464%11763.4-mic radius63%187971%114867%107410 nearest stores6%223770%132563%13655 nearest stores6%223770%132557%6962 nearest stores6%223770%132557%69670232770%132557%6967020203770%132557%6967020203770%132557%15815 nearest stores78%240575%143083%157220-mic radius69%218971%130649%137120-mic radius68%196273%114453%107410 nearest stores78%242275%132561%13655-mic radius68%196273%114453%107410 nearest stores78%242275%132561%13655-mic radius68%196273%114453%107410 nearest stores73%19414013613655-mic radius69%196273%114453%137510 nearest stores73%194140136	County	69%	2395	74%	1427	76%	1572
1 O-mile radius66%218973%130668%13715-mile radius67%204373%124668%12573.4-mile radius66%196273%114467%10742.5-mile radius63%187971%114857%107410 nearest stores76%223770%132556%11372 nearest stores64%223770%132556%11372 nearest stores64%223770%132556%11372 nearest stores64%223770%132557%1572 nearest stores73%240581%143076%15815 state73%240575%142775%15720-mile radius69%242270%132552%144910-mile radius69%242270%132552%144910-mile radius69%196273%119454%12573-mile radius68%196273%132561%13655 marest stores64%223775%132561%13655 marest stores64%242275%132561%13655 marest stores72%242275%132561%13655 marest stores64%2457100%1430100%15815 marest stores100%2405100%1430100%1581 </td <td>20-mile radius</td> <td>69%</td> <td>2242</td> <td>71%</td> <td>1325</td> <td>69%</td> <td>1449</td>	20-mile radius	69%	2242	71%	1325	69%	1449
5-mile radius 67% 2043 73% 1246 68% 1257 3.4-mile radius 66% 1962 73% 1194 64% 1176 2.5-mile radius 63% 1237 71% 1148 67% 1074 10 nearest stores 64% 2237 70% 1325 63% 1365 2 nearest stores 68% 2237 70% 1325 57% 66% 1367 2 nearest stores 68% 2237 70% 1325 57% 66% 1581 5 nearest stores 68% 2057 75% 1430 83% 1581 5 nearest stores 79% 2405 75% 1427 75% 1572 20-mile radius 69% 2424 71% 1325 52% 1449 10-mile radius 69% 2424 71% 1325 61% 1371 5 mile radius 68% 1879 71% 1246 54% 1257	10-mile radius	66%	2189	73%	1306	68%	1371
3.4-mile radius 66% 1962 73% 1194 64% 1176 2.5-mile radius 63% 1879 71% 1148 57% 1074 10 nearest stores 64% 2247 76% 1325 63% 1365 5 nearest stores 64% 2237 70% 1325 57% 696 Parent C. Centar stores within respondent: 68% 2405 81% 1430 83% 1581 County 71% 2405 75% 1430 76% 1572 20-mile radius 69% 2422 70% 1325 52% 1449 10-mile radius 69% 2424 70% 1325 52% 1449 10-mile radius 68% 1662 73% 1142 54% 1257 3-mile radius 68% 1692 73% 1148 53% 1074 10-mile radius 68% 1692 75% 1325 61% 1365 2	5-mile radius	67%	2043	73%	1246	68%	1257
2.5-mile radius 63% 1879 71% 1148 57% 1074 10 nearest stores 70% 2242 76% 1325 63% 1365 5 nearest stores 64% 2237 70% 1325 56% 137 2 nearest stores 68% 2337 70% 1325 57% 696 Panel C. Median TFP cost at stores within respondent: 73% 2405 75% 1430 76% 1581 County 71% 2395 72% 1427 75% 1572 20-mile radius 68% 2189 71% 1306 49% 1371 5-mile radius 67% 2043 71% 1346 54% 1257 3.4-mile radius 68% 1879 71% 1346 54% 1267 3.4-mile radius 68% 1879 71% 148 53% 1074 10 nearest stores 72% 2242 75% 1325 55% 1365 <td>3.4-mile radius</td> <td>66%</td> <td>1962</td> <td>73%</td> <td>1194</td> <td>64%</td> <td>1176</td>	3.4-mile radius	66%	1962	73%	1194	64%	1176
10 nearest stores 70% 2242 76% 1325 63% 1365 5 nearest stores 64% 2237 70% 1325 56% 1137 2 nearest stores 68% 2237 70% 1325 56% 696 Parel C. Median TFP cost at stores within respondents: 56% 1581 State 73% 2405 75% 1430 83% 1581 County 71% 2395 72% 1427 75% 1572 20-mile radius 69% 2242 70% 1325 52% 1449 10-mile radius 68% 1962 73% 1194 54% 1257 3.4-mile radius 68% 1962 73% 1345 53% 1365 5-mile radius 68% 1962 73% 1345 53% 1365 2.5-mile radius 68% 1962 73% 1325 61% 1365 5 nearest stores	2.5-mile radius	63%	1879	71%	1148	57%	1074
5 nearest stores 64% 2237 70% 1325 56% 1137 2 nearest stores 68% 2237 70% 1325 57% 696 Panel C. Median TPP cost at stores within respondents: 57% 696 Panel C. Median TPP cost at stores within respondents: 581 686 181% 1430 83% 1581 County 73% 2405 75% 1430 76% 1581 County 71% 2395 72% 1427 75% 1430 10-mile radius 68% 2189 71% 1326 54% 1257 3.4-mile radius 68% 1962 73% 1194 54% 1257 3.4-mile radius 68% 1962 73% 1325 61% 1365 2.5-mile radius 68% 1962 73% 1325 61% 1365 2.5-mile radius 100% 2037 63% 1325 61% 1365 <tr< td=""><td>10 nearest stores</td><td>70%</td><td>2242</td><td>76%</td><td>1325</td><td>63%</td><td>1365</td></tr<>	10 nearest stores	70%	2242	76%	1325	63%	1365
2 nearest stores68%223770%132557%696Paral C. Median TFP cost at stores within respondent:Census region78%240581%143083%1581State73%240575%143076%1581County71%239572%142775%147220-mile radius69%224270%132552%144910-mile radius68%196273%119454%12573.4-mile radius68%196273%119454%12673.4-mile radius68%196273%132561%13612.5-mile radius68%196273%132561%13673.4-mile radius68%196273%132561%13655 nearest stores72%224275%132561%13655 nearest stores72%224275%132561%13655 nearest stores72%23410%143010%1581County100%2405100%1430100%15816 county100%2395100%1430100%15816 county100%2395100%1430100%15816 county100%2395100%1430100%15817 county100%2395100%143090%14149 county100%2495<	5 nearest stores	64%	2237	70%	1325	56%	1137
Parel C. Median TPP cost at stores within respondent: Vertical stores within respondent: Census region 78% 2405 81% 1430 83% 1581 State 73% 2405 75% 1430 76% 1581 County 71% 2395 72% 1427 75% 1572 20-mile radius 69% 2242 70% 1325 52% 1449 10-mile radius 68% 2189 71% 1306 49% 1371 5-mile radius 67% 2043 71% 1246 54% 1257 3.4-mile radius 68% 1879 71% 1148 53% 1074 10 nearest stores 72% 2242 75% 1325 61% 1367 5 nearest stores 64% 2327 68% 1325 51% 1367 5 nearest stores 100% 2405 100% 1430 100% 1581 State 100% 2405 100%	2 nearest stores	68%	2237	70%	1325	57%	696
Census region78%240581%143083%1581State73%240575%143076%1581County71%239572%142775%157220-mile radius69%224270%132552%144910-mile radius68%218971%130649%13715-mile radius68%18971%124654%12573.4-mile radius68%187971%114853%107410 nearest stores72%224275%132561%13555 nearest stores72%224275%132561%13655 nearest stores64%23768%132555%1365Census region100%2405100%1430100%15815 nearest stores100%2405100%1430100%15816 County100%2405100%1427100%15816 County100%2405100%1427100%157210 -mile radius100%2489100%132597%13715 -mile radius99%2424100%132697%13715 -mile radius99%248100%1449144910 -mile radius99%248100%132597%13715 -mile radius97%187999%124697%13716 -d	Panel C. Median TFP cost at stores within respondent's						
State 73% 2405 75% 1430 76% 1581 County 71% 2395 72% 1427 75% 1572 20-mile radius 69% 2242 70% 1325 52% 1449 10-mile radius 68% 2189 71% 1306 49% 1371 5-mile radius 67% 2043 71% 1246 54% 1257 3.4-mile radius 68% 1962 73% 1194 54% 1257 2.5-mile radius 68% 1962 73% 1148 53% 1074 2.5-mile radius 68% 1879 71% 1148 53% 1074 2.5-mile radius 64% 2237 68% 1325 55% 1137 Panel D. Minimum TFP cost at stores within respondent/s 100% 1430 100% 1581 State 100% 2405 100% 1430 100% 1572 20-mile radius	Census region	78%	2405	81%	1430	83%	1581
County71%239572%142775%157220-mile radius69%224270%132552%144910-mile radius68%218971%130649%13715-mile radius67%204371%124654%12573.4-mile radius68%196273%119454%11762.5-mile radius68%187971%114853%107410 nearest stores72%224275%132561%13655 nearest stores60%20768%132561%13657 no mart TP cost at stores within responter75%132561%1581County100%2405100%1430100%1581County100%2395100%1325100%144910-mile radius99%2242100%1325100%144910-mile radius98%1962100%134697%12572.5-mile radius98%1962100%1325100%144910-mile radius98%1962100%134697%12572.5-mile radius98%1962100%124697%12573.4-mile radius98%1962100%134697%12573.4-mile radius98%1962100%124697%12573.5-mile radius98%1962100%124697%1257 <td>State</td> <td>73%</td> <td>2405</td> <td>75%</td> <td>1430</td> <td>76%</td> <td>1581</td>	State	73%	2405	75%	1430	76%	1581
20-mic69%224270%132552%144910-mile radius68%218971%130649%13715-mile radius67%204371%124654%12573.4-mile radius68%196273%119454%11762.5-mile radius68%187971%114853%107410 nearest stores72%224275%132561%13655 nearest stores64%203768%132555%1325Parel D. Minimum TFP cost at stores within responderts:Census region100%2405100%1430100%1581County100%2395100%1427100%157220-mile radius99%2242100%1325100%144910-mile radius99%2242100%132697%13715-mile radius98%204399%124697%12573.4-mile radius98%1962100%114881%107410 nearest stores98%1962100%132581%10742.5-mile radius97%187999%114881%107410 nearest stores99%2242100%132581%13655 nearest stores99%2242100%132560%13255 nearest stores99%2242100%132560%1365<	County	71%	2395	72%	1427	75%	1572
10-mile radius 68% 2189 71% 1306 49% 1371 5-mile radius 67% 2043 71% 1246 54% 1257 3.4-mile radius 68% 1962 73% 1194 54% 1176 2.5-mile radius 68% 1962 73% 1148 53% 1074 10 nearest stores 72% 2242 75% 1325 61% 1365 5 nearest stores 64% 237 68% 1325 55% 1137 Parel D. Minimum TFP cost at stores within respondenter Census region 100% 2405 100% 1430 100% 1581 State 100% 2395 100% 1427 100% 1572 20-mile radius 99% 2422 100% 1325 100% 1449 10-mile radius 100% 2395 100% 1427 100% 1572 20-mile radius 99% 2442 100% 1325 100% 1371 5-mile radius 98% 1962 100%	20-mile radius	69%	2242	70%	1325	52%	1449
5-mile radius67%204371%124654%12573.4-mile radius68%196273%119454%11762.5-mile radius68%187971%114853%107410 nearest stores72%224275%132561%13655 nearest stores64%22768%13255%1376Parel D. Minimum TFP cost at stores within respondent/Parel D. Minimum TFP cost at stores within respondent/VCensus region100%2405100%1430100%158160%209%100%1430100%157220-mile radius99%2242100%1325100%144910-mile radius99%2242100%130697%12573.4-mile radius98%204399%124697%12573.4-mile radius98%1962100%119489%11762.5-mile radius97%187999%114881%107410 nearest stores99%1245100%1325107410 nearest stores97%223798%132567%11372 nearest stores97%223784%132559%696	10-mile radius	68%	2189	71%	1306	49%	1371
3.4-mile radius 68% 1962 73% 1194 54% 1176 2.5-mile radius 68% 1879 71% 1148 53% 1074 10 nearest stores 72% 2242 75% 1325 61% 1365 5 nearest stores 64% 2237 68% 1325 55% 1137 Pamel D. Minimum TFP cost at stores within respondent: Census region 100% 2405 100% 1430 100% 1581 State 100% 2405 100% 1430 100% 1572 20-mile radius 100% 2495 100% 1427 100% 1572 20-mile radius 99% 2242 100% 1325 100% 1449 10-mile radius 98% 2043 99% 1246 97% 1257 3.4-mile radius 98% 1962 100% 1194 89% 1176 2.5-mile radius 98% 1962 100% 1194 89% 1176 2.5-mile radius 98% 1962	5-mile radius	67%	2043	71%	1246	54%	1257
2.5-mile radius 68% 1879 71% 1148 53% 1074 10 nearest stores 72% 2242 75% 1325 61% 1365 5 nearest stores 64% 2237 68% 1325 55% 1137 Panel D. Minimum TFP cost at stores within respondents: V Census region 100% 2405 100% 1430 100% 1581 State 100% 2405 100% 1430 100% 1581 County 100% 2395 100% 1430 100% 1581 10-mile radius 99% 2242 100% 1325 100% 1449 10-mile radius 99% 2242 100% 1325 100% 1449 10-mile radius 98% 2043 99% 1246 97% 1257 3.4-mile radius 98% 1962 100% 1194 89% 1176 2.5-mile radius 98% 2242 100% 1325 81% 1074 10 nearest stores 9	3.4-mile radius	68%	1962	73%	1194	54%	1176
10 nearest stores 72% 2242 75% 1325 61% 1365 5 nearest stores 64% 2237 68% 1325 55% 1137 Panel D. Minimum TFP cost at stores within respontent: 1430 100% 1581 State 100% 2405 100% 1430 100% 1581 County 100% 2405 100% 1430 100% 1581 County 100% 2395 100% 1427 100% 1572 20-mile radius 99% 2242 100% 1325 100% 1449 10-mile radius 90% 2189 100% 1326 97% 1371 5-mile radius 98% 2043 99% 1246 97% 1257 3.4-mile radius 98% 1962 100% 1194 89% 1176 2.5-mile radius 97% 1879 99% 1148 81% 1074	2.5-mile radius	68%	1879	71%	1148	53%	1074
5 nearest stores 64% 2237 68% 1325 55% 1137 Panel D. Minimum TFP cost at stores within respondents: Census region 100% 2405 100% 1430 100% 1581 State 100% 2405 100% 1430 100% 1581 County 100% 2395 100% 1427 100% 1572 20-mile radius 99% 2242 100% 1325 100% 1449 10-mile radius 100% 2189 100% 1306 97% 1371 5-mile radius 98% 2043 99% 1246 97% 1257 3.4-mile radius 98% 1962 100% 1194 89% 1176 2.5-mile radius 98% 1962 100% 1188 81% 1074 2.5-mile radius 97% 1879 99% 1448 81% 1074 2.5-mile radius 97% 2242 100% 1325 <t< td=""><td>10 nearest stores</td><td>72%</td><td>2242</td><td>75%</td><td>1325</td><td>61%</td><td>1365</td></t<>	10 nearest stores	72%	2242	75%	1325	61%	1365
Panel D. Minimum TFP cost at stores within respondent's: Census region 100% 2405 100% 1430 100% 1581 State 100% 2405 100% 1430 100% 1581 County 100% 2395 100% 1427 100% 1572 20-mile radius 99% 2242 100% 1325 100% 1449 10-mile radius 100% 2189 100% 1306 97% 1371 5-mile radius 98% 2043 99% 1246 97% 1257 3.4-mile radius 98% 1962 100% 1194 89% 1176 2.5-mile radius 98% 1962 100% 1148 81% 1074 4.5-mile radius 97% 1879 99% 1448 1365 1365 2.5-mile radius 97% 2242 100% 1325 81% 1365 10 nearest stores 99% 2242 100% 1325 67% <td>5 nearest stores</td> <td>64%</td> <td>2237</td> <td>68%</td> <td>1325</td> <td>55%</td> <td>1137</td>	5 nearest stores	64%	2237	68%	1325	55%	1137
Census region100%2405100%1430100%1581State100%2405100%1430100%1581County100%2395100%1427100%157220-mile radius99%2242100%1325100%144910-mile radius100%2189100%130697%13715-mile radius98%204399%124697%12573.4-mile radius98%1962100%119489%11762.5-mile radius97%187999%114881%107410 nearest stores99%2242100%132581%13655 nearest stores97%223784%132559%696	Panel D. Minimum TFP cost at stores within responder	ıt's:					
State100%2405100%1430100%1581County100%2395100%1427100%157220-mile radius99%2242100%1325100%144910-mile radius100%2189100%130697%13715-mile radius98%204399%124697%12573.4-mile radius98%1962100%119489%11762.5-mile radius97%187999%114881%107410 nearest stores99%2242100%132581%13655 nearest stores97%223784%132559%696	Census region	100%	2405	100%	1430	100%	1581
County100%2395100%1427100%157220-mile radius99%2242100%1325100%144910-mile radius100%2189100%130697%13715-mile radius98%204399%124697%12573.4-mile radius98%1962100%119489%11762.5-mile radius97%187999%148881%107410 nearest stores99%2242100%132581%13655 nearest stores97%223784%132559%696	State	100%	2405	100%	1430	100%	1581
20-mile radius99%2242100%1325100%144910-mile radius100%2189100%130697%13715-mile radius98%204399%124697%12573.4-mile radius98%1962100%119489%11762.5-mile radius97%187999%114881%107410 nearest stores99%2242100%132581%13655 nearest stores97%223798%132559%696	County	100%	2395	100%	1427	100%	1572
10-mile radius100%2189100%130697%13715-mile radius98%204399%124697%12573.4-mile radius98%1962100%119489%11762.5-mile radius97%187999%114881%107410 nearest stores99%2242100%132581%13655 nearest stores97%223784%132559%696	20-mile radius	99%	2242	100%	1325	100%	1449
5-mile radius98%204399%124697%12573.4-mile radius98%1962100%119489%11762.5-mile radius97%187999%114881%107410 nearest stores99%2242100%132581%13655 nearest stores97%223798%132567%11372 nearest stores82%223784%132559%696	10-mile radius	100%	2189	100%	1306	97%	1371
3.4-mile radius98%1962100%119489%11762.5-mile radius97%187999%114881%107410 nearest stores99%2242100%132581%13655 nearest stores97%223798%132567%11372 nearest stores82%223784%132559%696	5-mile radius	98%	2043	99%	1246	97%	1257
2.5-mile radius97%187999%114881%107410 nearest stores99%2242100%132581%13655 nearest stores97%223798%132567%11372 nearest stores82%223784%132559%696	3.4-mile radius	98%	1962	100%	1194	89%	1176
10 nearest stores 99% 2242 100% 1325 81% 1365 5 nearest stores 97% 2237 98% 1325 67% 1137 2 nearest stores 82% 2237 84% 1325 59% 696	2.5-mile radius	97%	1879	99%	1148	81%	1074
5 nearest stores 97% 2237 98% 1325 67% 1137 2 nearest stores 82% 2237 84% 1325 59% 696	10 nearest stores	99%	2242	100%	1325	81%	1365
2 nearest stores 82% 2237 84% 1325 59% 696	5 nearest stores	97%	2237	98%	1325	67%	1137
	2 nearest stores	82%	2237	84%	1325	59%	696

Notes: For each sample, the first column displays the survey-weighted fraction of households who can afford the local cost of the TFP. The local cost of the TFP is estimated as described in Table 2. Eligibility for SNAP is estimated in FoodAPS using the MATH SIPP + approach (Leftin et al., 2014); we use Model 4. See text for further discussion.

shortfalls that are generally smaller.

These average shortfalls are large relative to households' SNAP benefits and incomes. For the sake of illustration, consider SNAP-recipient households who cannot afford the TFP at median county-level prices. On average, such households face a dollar shortfall of \$160, receive \$235 per month in benefits, and have net income of \$557 per month. Therefore, the \$160 difference between the local TFP cost and their available resources for purchasing food is approximately 68 percent of benefits, or 29 percent of net income.

3.5. Limitations of our analysis

Our analysis is impacted by two main data limitations, which affect how our results should be interpreted. First, we study SNAP purchasing power with respect to local variation in food prices only, without controlling for local differences in prices of other goods. The one exception is our analysis which uses self-reported SNAP benefits plus 30 percent of net income as the measure of resources available to spend on food. Here, some of the deductions that affect a household's net income (e.g., the excess shelter deduction and deductions for child care costs and medical expenditures for the elderly) will vary with local costs of non-food goods. Beyond this, however, we are unable to control for other costs of living at such a fine geographic level, and it may be that other costs do not follow the same pattern of food cost differences across geography. If local non-food prices are lower where food prices are high, then low SNAP purchasing power is offset by higher purchasing power with respect to other goods. On the other hand, if local non-food prices are positively correlated with food prices across areas, the picture for SNAP households may be more dismal than the one we have painted.

Second, our study focuses on food costs primarily in urban areas because the FoodAPS SNAP sample is comprised almost entirely of households who live in metropolitan areas (see Table 3). Only about 10 percent of the SNAP households in our sample reside in non-

G. Christensen and E.T. Bronchetti

Table 5

Size of monthly shortfall for SNAP recipient households who cannot afford the local cost of the TFP.

		SNAP Benefits + 3 Insufficient to Cover	80% Net Income r Local TFP Cost	Maximum SNAP Benefit Insufficient to Cover Local TFP Cost	
Measure of TFP Co	ost	Average Dollar Amt. of Shortfall	Sample Size	Average Dollar Amt. of Shortfall	Sample Size
Panel A					
1 unet 71.	Statutory TFP Cost (\$145)	\$155	391	_	0
	Mean TFP Cost of all Stores in IRI Data (\$156)	\$150	511	\$2	1581
	TFP Cost. Nearest Store to Respondent	\$170	344	\$68	433
	TFP Cost, Primary Store where Respondent Reports Shopping	\$133	203	\$44	257
	TFP Cost, Alternate Store where Respondent Reports Shopping	\$176	123	\$53	157
	Mean TFP Cost, Primary and Alternate Stores	\$141	264	\$45	283
	TFP Cost, Store where Respondent Shopped Most	\$136	172	\$43	223
Panel B. Mean TFF	cost at stores within respondent's:				
	Census region	\$167	488	\$45	554
	State	\$177	434	\$39	399
	County	\$162	425	\$55	418
	20-mile radius	\$174	393	\$55	428
	10-mile radius	\$173	389	\$62	396
	5-mile radius	\$168	377	\$67	371
	3.4-mile radius	\$172	363	\$67	345
	2.5-mile radius	\$173	345	\$68	363
	10 nearest stores	\$172	394	\$92	339
	5 nearest stores	\$164	408	\$84	415
	2 nearest stores	\$159	386	\$70	418
Panel C. Median T	FP cost at stores within respondent's:				
	Census region	\$163	440	\$12	353
	State	\$161	422	\$18	406
	County	\$160	394	\$20	382
	20-mile radius	\$157	374	\$19	385
	10-mile radius	\$158	362	\$21	381
	5-mile radius	\$164	354	\$30	378
	3.4-mile radius	\$162	344	\$29	323
	2.5-mile radius	\$159	322	\$23	340
	10 nearest stores	\$151	366	\$26	333
	5 nearest stores	\$145	377	\$29	432
Panel D. Minimum	TFP cost at stores within respondent's:				
	Census region	-	0	-	0
	State	\$77	21	-	0
	County	\$95	80	\$148	6
	20-mile radius	\$99	67	-	0
	10-mile radius	\$107	93	\$57	4
	5-mile radius	\$105	115	\$102	12
	3.4-mile radius	\$114	124	\$71	13
	2.5-mile radius	\$112	131	\$32	14
	10 nearest stores	\$105	149	\$15	2
	5 nearest stores	\$131	191	\$27	28
	2 nearest stores	\$159	276	\$51	207

Notes: Table contains the survey-weighted average shortfall between SNAP benefits plus 30 percent of net income (or max benefits for household size) and local TFP cost. Local cost of the TFP is estimated as described in Table 2. Sample is SNAP-recipient households for whom the shortfall is greater than zero (i.e., resources are insufficient to purchase the full TFP).

metropolitan areas, and SNAP purchasing power is likely to be higher in rural areas, suggesting that our estimates understate the extent to which SNAP benefits allow households to afford the TFP overall. That said, other differences between urban and rural areas may offset disparities in SNAP purchasing power, like differences in access to food, the availability of community/non-profit food assistance, and other nonfood costs (e.g., transportation). Without a fuller picture of these differences, we caution readers to interpret our results as reflecting SNAP purchasing power for a sample of primarily urban households.

4. Policy implications

From a policy standpoint, it is crucial to consider how large these shortfalls are in aggregate. That is, what would it cost to enable 100 percent of SNAP households to afford the TFP? Would doing so be possible through redistribution, or would it require additional program spending? We explore these questions in Figs. 2 and 3 which display the distribution of budget shortfalls for *all* SNAP recipient households. Fig. 2 displays these shortfalls when households are assumed to be able to spend their SNAP benefit plus 30 percent of net income, and Fig. 3 shows shortfalls calculated using maximum SNAP benefits as the relevant measure of household resources.

The shortfall distributions are centered around small negative amounts, where negative amounts reflect *high* SNAP purchasing power, i.e., that household resources are more than sufficient to purchase the TFP. Summing this difference across all SNAP households provides a large negative number on the order of \$3 to \$5 billion. (These sums are shown in Appendix Table 6) This implies that if it were costless to redistribute some benefit dollars from households who are more than able to afford the TFP in their areas to those who are unable to do so, current levels of program funding and total benefit payments would be adequate to enable every recipient to purchase the TFP locally.



Fig. 2. Distribution of budget shortfalls. (Difference between local TFP cost and SNAP benefit + 30 percent of net income).



Fig. 3. Distribution of budget shortfalls. (Difference between local TFP cost and maximum SNAP benefit).

An important implication of our results is that indexing benefits according to local food prices would be one way to achieve such redistribution. Of course, adjusting SNAP benefits for geographic variation in food prices is easier said than done. Historically, a lack of data on local food prices has prevented policymakers from considering direct adjustments to SNAP benefits to account for geographic differences in the cost of food. While the Bureau of Labor Statistics (BLS) provides regional CPIs for the four major Census regions and for 27 metropolitan statistical areas (MSAs), the regional price indices are meant to reflect changes in prices within regions over time, not to compare prices across areas at a given point in time (Bureau of Labor Statistics, 2018). On the other hand, it seems possible that Nielsen and IRI scanner data could be used by the USDA to calculate regional food price indices. The USDA already has access to these proprietary data, and Nielsen data have been used by Feeding America to construct a relative price index at the county level for its Map the Meal Gap project (Feeding America 2018).

Of course, we note that adjusting benefits to geographic differences in food prices would mean lowering the maximum benefit in many areas (those with lower food prices), which is prevented by current law. As we describe above, doing so would also presumably lower benefits in rural areas, where access to other forms of food assistance may be lower and non-food costs like transportation costs may be higher. Rather, a permissible – but admittedly more costly – adjustment would be to leave benefits as they are for low- or average-cost areas, and raise the maximum benefit only in areas with higher-than-average food prices (Institute of Medicine and National Research Council, 2013).²⁵ Such a policy would cost an additional \$83 million per month in benefits, or just under \$1 billion per year. Compared to the program's \$75 billion in benefit payouts in 2012, this reflects an increase in costs of approximately 1 percent.

Table 6 illustrates such an approach, supposing that SNAP households are given benefits equal to the larger of their current legislated benefit or the median cost of the TFP in their county. This adjustment raises the fraction of SNAP households who can afford the local cost of the TFP substantially, often by more than 10 percentage points. For example, when shoppers are assumed to purchase the TFP at the median price in a 3.4-mile radius, we now find that 86 percent of SNAP households can afford the TFP, compared to just 74 percent under current benefits (see Table 2). We estimate that 26 percent of SNAP households would receive increased benefits; the average increase (among those receiving increases) would be just under \$20 per month. However, we note that a substantial fraction of households - in the range of 12 to 16 percent - still cannot afford the local cost of the TFP when they are given benefits equal to the median TFP price in their county. This reflects the fact that SNAP households reside in particularly high-priced pockets of their counties. Thus, while adjusting benefits for high county food prices goes a long way toward making the TFP affordable to more SNAP households, such a policy would still leave some households unable to afford the TFP.

5. Conclusions

This study provides new descriptive evidence on the adequacy of SNAP benefits to purchase a low-cost, nutritious diet as specified by the Thrifty Food Plan (TFP), which is the basis for legislated SNAP benefit levels. Acknowledging that a given amount of SNAP benefits will buy less food in areas with high food prices, we estimate the fraction of SNAP households that are able to purchase the TFP at local prices (i.e., the "sufficiency rate"). Using FoodAPS data to answer this question, we account for geographic variation in local food prices in much finer

Table 6

Fraction of SNAP households who can afford the local cost of the TFP with benefits adjusted for high local food prices. (Household's simulated SNAP benefit set equal to greater of legislated max benefit or county-level median TFP price).

	Simulated SNAP Benefit Exceeds Local TFP Cost?			
Measure of TFP Cost	% of Households	Sample Size		
Median TFP cost at stores within respondent's:				
County	100%	1581		
20-mile radius	88%	1467		
10-mile radius	87%	1442		
5-mile radius	86%	1347		
3.4-mile radius	86%	1290		
2.5-mile radius	84%	1234		
Mean TFP cost at stores wit	hin respondent's:			
County	100%	1581		
20-mile radius	83%	1467		
10-mile radius	84%	1442		
5-mile radius	84%	1347		
3.4-mile radius	83%	1290		
2.5-mile radius	80%	1234		

detail than has previously been possible.

Our main findings indicate that a substantial share of SNAP-recipient households - on the order of 20-25 percent - face local TFP prices that are too high to purchase the TFP with SNAP benefits plus 30 percent of net income. This share increases only slightly as we expand the distance within which the household is assumed to be able to shop. For households who are unable to afford the TFP, average dollar shortfalls between the cost of the TFP and SNAP benefits plus 30 percent of income are often as large as \$150 per month. When SNAP recipients are assumed to be able to purchase the TFP at the minimum-cost store in 10-20 miles, SNAP benefits are sufficient for a much larger fraction of recipient households (90-95 percent). However, the share who can afford the TFP is still less than 100 percent, and the assumption that households are able to identify and travel to the minimum TFP-cost store in a large area seems particularly unlikely for this population. Only 67 percent of SNAP recipient households in our sample have a car, and the vast majority live in metropolitan areas, where the costs of such travel might be prohibitively high. Importantly, we find that sufficiency rates of 100 percent could be achieved without additions to total benefit payouts by redistributing some benefit dollars from those for whom SNAP is more than sufficient to purchase the local TFP to those households who are currently unable to afford their local TFP. However, such a redistribution is easier said than done. In fact, we demonstrate that even a policy that provided benefits equal to the maximum of the current legislated benefit or the county-level TFP price would not make the local cost of the TFP affordable for all SNAP households. Because SNAP households reside in particularly expensive parts of their counties, we find that 12 to16 percent of SNAP households are still unable to afford the TFP under such a policy.²⁶

Finally, our focus on geographic variation in food prices and in the *real* value of SNAP benefits also suggests a new avenue for research on the food stamp program and its impacts. That legislated SNAP benefit levels are set at the national level presents challenges for quasi-experimental analysis of the causal impacts of SNAP on outcomes of interest. Geographic variation in food prices, however, presents a plausibly exogenous source of variation in SNAP generosity (in real terms), that researchers can use to study the effects of the program on outcomes like health (Bronchetti, Christensen, and Hoynes 2019), food security

²⁵ See Institute of Medicine and National Research Council (2013), Chapters 2 and 5 for details on the 1975 court decision, and eventual rewriting of the law, that make the TFP the basis for benefits, as well as information on the difficulties of geographic adjustment of SNAP benefits.

 $^{^{26}}$ This simulation assumes that households face a local TFP price ranging from the median cost of the TFP within a 2.5-mile radius (in which case, 16 percent are unable to afford it) to the median TFP cost at stores within a 20-mile radius (in which case, 12 percent cannot afford the TFP).

G. Christensen and E.T. Bronchetti

(Gregory and Coleman-Jensen 2013), or nutrition (Bronchetti, Christensen, and Hansen 2017).

6. Author note

This project was supported with a grant from the University of Kentucky Center for Poverty Research through funding by the U.S. Department of Agriculture, Economic Research Service and the Food and Nutrition Service, Agreement Numbers 58-5000-1-0050 and 58-5000-3-0066. The opinions and conclusions expressed herein are solely those of the author(s) and should not be construed as representing the opinions or policies of the sponsoring agencies.

Any opinions and conclusions expressed herein are those of the authors and do not necessarily reflect the views of the U.S. Census Bureau. All results have been reviewed to ensure that no confidential information is disclosed.

We are grateful to USDA-ERS and to UKCPR for generous funding, to Benjamin Hansen, Craig Gundersen, Linlin Fan, and Michele ver Ploeg for suggestions, and to staff at NORC and USDA-ERS for support with the FoodAPS data.

CRediT authorship contribution statement

Garret Christensen: Funding acquisition, Conceptualization, Software, Visualization, Formal analysis. **Erin Todd Bronchetti:** Funding acquisition, Conceptualization, Software, Visualization, Formal analysis.

Appendix A. Supplementary material

Supplementary data to this article can be found online at https://doi.org/10.1016/j.foodpol.2020.101937.

References

- Albouy, D., 2009. The unequal geographic burden of federal taxation. J. Polit. Econ. 117 (4), 635–667. https://doi.org/10.1086/605309.
- Almond, D., Hoynes, H.W., Schanzenbach, D.W., 2011. Inside the war on poverty: the impact of food stamps on birth outcomes. Rev. Econ. Stat. 93 (2), 387–403. https:// doi.org/10.1162/REST_a_00089.
- Bronchetti, Erin, Garret Christensen, Benjamin Hansen, 2017. Local Food Prices and Nutrition in SNAP Households. In: Presented at the APPAM Fall Research Conference, Chicago, IL, November 3. https://github.com/garretchristensen/ FoodAPSAPPAM2017.
- Bronchetti, E., Christensen, G., Hoynes, H., 2019. Local food prices, SNAP purchasing power, and child health. J. Health Econ. 68, 1022–1031. https://doi.org/10.1016/j jhealeco.2019.102231.
- Bureau of Labor Statistics, 2018. Consumer price index frequently asked questions: U.S. Bureau of labor statistics. Consumer Price Index Frequently Asked Questions. April 13, 2018. https://www.bls.gov/cpi/questions-and-answers.htm#Question_20.
- Çakır, M., Beatty, T.K.M., Boland, M.A., Park, T.A., Snyder, S., Wang, Y., 2018. Spatial and Temporal Variation in the Value of the Women, Infants, and Children Program's Fruit and Vegetable Voucher. Am. J. Agric. Econ. 100 (3), 691–706. https://doi.org/ 10.1093/ajae/aay002.
- Carlson, Andrea, Lino, Mark, Juan, Wen Yen, Hanson, Kenneth, Basiotis, P. Peter, 2007. Thrifty Food Plan, 2006." CNPP-19. US Department of Agriculture, Center for Nutrition Policy and Promotion. http://www.cnpp.usda.gov/sites/default/files/ usda_food_plans_cost_of_food/TFP2006Report.pdf.
- Clay, Marie, Ver Ploeg, Michele, Coleman-Jensen, Alisha, Elitzak, Howard, Gregory, Christian A., Levin, David, Newman, Constance, Rabbitt, Matthew P., 2016. Comparing National Household Food Acquisition and Purchase Survey (FoodAPS) Data With Other National Food Surveys' Data. EIB-157. Economic Research Service.
- Coleman-Jensen, Alisha, Nord, Mark, Andrews, Margaret, Carlson, Steven, 2012. Household food security in the United States in 2011. United States Department of Agriculture, Economic Research Service. https://www.ers.usda.gov/webdocs/ publications/45020/30967_err141.pdf?v=41165.
- DellaVigna, S., Gentzkow, M., 2019. Uniform pricing in U. S. Retail Chains. Quart. J. Econ. 134 (4), 2011–2084. https://doi.org/10.1093/qje/qjz019.
- East, Chloe N., 2016. The Effect of Food Stamps on Children's Health: Evidence from

Immigrants' Changing Eligibility." University of Colorado Denver. http://cneast. weebly.com/uploads/8/9/9/7/8997263/east_jmp.pdf.

- Fan, Linlin, Baylis, Kathy, Gundersen, Craig, ver Ploeg, Michelle, 2018. Does a nutritious diet cost more in food deserts? Agric. Econ. 49(5), 587–597.
- Feeding America, 2018. Map the Meal Gap: A Report on County and Congressional District Food Insecurity and County Food Cost in the United States in 2016. https:// www.feedingamerica.org/sites/default/files/research/map-the-meal-gap/2016/ 2016-map-the-meal-gap-full.pdf.
- Gregory, Christian A., Coleman-Jensen, Alisha, 2013. Do high food prices increase food insecurity in the United States? Appl. Econ. Perspect. Policy 35 (4), 679–707. https:// doi.org/10.1093/aepp/ppt024.
- Gregory, Christian A., Deb, Partha, 2015. Does SNAP improve your health? Food Policy 50 (January), 11–19. https://doi.org/10.1016/j.foodpol.2014.09.010.
- Gregory, Christian, Rabbitt, Matthew P., Ribar, David C., 2015. The supplemental nutrition assistance program and food insecurity. In: Bartfeld, Judith, Gundersen, Craig, Smeeding, Timothy, Ziliak, James P. (Eds.), SNAP Matters: How Food Stamps Affect Health and Well-Being. Stanford University Press, pp. 74–106.
- Gunderson, Craig, Baylis, Kathy, Fan, Linlin, House, Lisa, Dutko, Paula. 2016. Construction of weekly store-level food basket costs: documentation. https://www. ers.usda.gov/media/8582/priceindexdata.pdf.
- Hoynes, Hilary, Schanzenbach, Diane Whitmore, 2016. US Food and Nutrition Programs. In: Economics of Means-Tested Transfer Programs in the United States, Volume I. University Of Chicago Press. http://www.press.uchicago.edu/ucp/books/book/ chicago/E/bo23520704.html.
- Hoynes, Hilary, Schanzenbach, Diane Whitmore, Almond, Douglas, 2016. Long-run impacts of childhood access to the safety net. Am. Econ. Rev. 106 (4), 903–934. https:// doi.org/10.1257/aer.20130375.
- Leftin, Joshua, Smith, Joel, Cunnyngham, Karen, Trippe, Carole, 2014. Creation of the 2011 MATH SIPP + Microsimulation Model and Database. Technical Working Paper. https://www.mathematica-mpr.com/our-publications-and-findings/publications/ creation-of-the-2011-math-sipp-microsimulation-model-and-database.
- Leibtag, Ephraim, Kumcu, Aylin, 2011. The WIC fruit and vegetable cash voucher: does regional price variation affect buying power? 75. USDA ERS Economic Information Bulletin. http://www.ers.usda.gov/publications/pub-details/?puble = 44545.
- Muth, Mary K., Sweitzer, Megan, Brown, Dreick, Capogrossi, Kristen, Karns, Shawn, Levin, David, Okrent, Abigail, Siegel, Peter, Zhen, Chen, 2016. Understanding IRI household-based and store-based scanner data. 1942. USDA ERS Technical Bulletin. https://www.ers.usda.gov/webdocs/publications/47633/57105_tb-1942.pdf?v = 42473.
- Mykerezi, Elton, Mills, Bradford, 2010. The impact of food stamp program participation on household food insecurity. Am. J. Agric. Econ. 92 (5), 1379–1391.
- Nord, M., Golla, M., 2009. Does SNAP decrease food insecurity? Untangling the selfselection effect. USDA. United States Department of Agriculture, Economic Research Service.
- Schmeiser, Maximilian D., 2012. The impact of long-term participation in the supplemental nutrition assistance program on child obesity. Health Econ. 21 (4), 386–404. https://doi.org/10.1002/hec.1714.
- Schmidt, Lucie, Shore-Sheppard, Lara, Watson, Tara, 2016. The effect of safety-net programs on food insecurity. J. Human Resour. 51 (3), 589–614. https://doi.org/10. 3368/jhr.51.3.1013-5987R1.
- Shaefer, H. Luke, Gutierrez, Italo A., 2013. The supplemental nutrition assistance program and material hardships among low-income households with children. Social Service Rev. 87 (4), 753–779. https://doi.org/10.1086/673999.
- Supplemental Nutrition Assistance Program: Examining the Evidence to Define Benefit Adequacy, 2013. Institute of Medicine and National Research Council. http://www. iom.edu/Reports/2013/Supplemental-Nutrition-Assistance-Program-Examining-the-Evidence-to-Define-Benefit-Adequacy.aspx.
- Todd, Jessica E., Leibtag, Ephraim, Penberthy, Corttney, 2011. Geographic differences in the relative price of healthy foods. United States Department of Agriculture, Economic Research Service. http://books.google.com/books?hl=en&lr=&id= 4qm5sgn3u20C&oi=fnd&pg=PP5&dq=todd+qfahpd&ots=UPHWMF7gnM&sig= 6wOvj14JceZtjJ-d7S111pPJm2w.
- Todd, Jessica E., Mancino, Lisa, Leibtag, Ephraim S., Tripodo, Christina, 2010. Methodology behind the quarterly food-at-home price database. United States Department of Agriculture, Economic Research Service. http://ideas.repec.org/p/ ags/uerstb/97799.html.
- Ver Ploeg, Michele, Mancino, Lisa, Todd, Jessica E., Clay, Dawn Marie, Scharadin, Benjamin, 2015. Where Do Americans Usually Shop for Food and How Do They Travel To Get There? Initial Findings From the National Household Food Acquisition and Purchase Survey. EIB-138. U.S. Department of Agriculture, Economic Research Service. https://www.ers.usda.gov/webdocs/publications/43953/eib138_errata. pdf?y = 300.4.
- Wilde, Parke E., Llobrera, Joseph, 2009. Using the thrifty food plan to assess the cost of a nutritious diet. J. Consumer Affairs 43 (2), 274–304. https://doi.org/10.1111/j. 1745-6606.2009.01140.x.
- Yen, Steven T., Andrews, Margaret, Chen, Zhuo, Eastwood, David B., 2008. Food stamp program participation and food insecurity: an instrumental variables approach. Am. J. Agric. Econ. 90 (1), 117–132.